Author's response to reviews

Title: Identification of breast cancer patients with a high risk of developing brain metastases: A single-institutional retrospective analysis.

Authors:

Volker Rudat (volker.rudat@gmail.com)
Hamdan El-Sweilmeen (HEl-Sweilmeen@saad.com.sa)
Iris Brune-Erber (IErber@saadmedical.com)
Alaa A Nour (ANour@saadmedical.com)
Nidal Almasri (NMasri@saadmedical.com)
Saleh Altuwaijri (saltuwaijri@saadmedical.com)
Elias Fadel (EFadel@saadmedical.com)

Version: 4 Date: 2 April 2014

Author's response to reviews:

As requested we added the full name of the Institutional Review Board that provided Ethical approval should be included in the Methods section last sentence):

This study was approved by the local Institutional Review Board “Institutional Review Board - Saad Specialist Hospital (Registration number: H-05-KH-001, King Abdul-Aziz City of Science and Technology – KACST)” and performed in compliance with the Helsinki Declaration.

We did not add the section "Acknowledgements" because all contributions were done by the authors.